Author’s response to reviews

Title: Helicobacter pylori Infection and its Association with Anemia among Dyspeptic Patients Attending Butajira Hospital, Ethiopia

Authors:

Dargaze Kibru (kdargaze@yahoo.com)
Baye Gelaw (tedybayegelaw@gmail.com)
Agersew Alemu (agersewalemu@yahoo.com)
Zelalem Addis (zelalemaddis@ymail.com)

Version: 2  Date: 18 August 2014

Author’s response to reviews: see over
Covering letter: We the authors of this paper declare that this submitted version of the paper to BMC infectious diseases journal is up on our agreement and the material is original and neither published elsewhere nor submitted for publication simultaneously. We decide to submit this research article to this journal because initially we were advised to do this by the editors of infectious disease of poverty journal. But later we found that the work has direct relevance to the journal and also it is focus area. Moreover, we believe the article would add something to the scientific community when published on this journal.